Author's response to reviews

Title: Self-Reported Dental Hygiene and Health, Obesity, and Systemic Inflammation in a Pediatric Rural Community Cohort

Authors:

Stephanie J Frisbee (sfrisbee@hsc.wvu.edu)
Christopher B Chambers (cchambe5@mix.wvu.edu)
Jefferson C Frisbee (jfrisbee@hsc.wvu.edu)
Adam G Goodwill (agoodwill@hsc.wvu.edu)
Richard J Crout (rcrout@hsc.wvu.edu)

Version: 2 Date: 25 August 2009

Author’s response to reviews:

August 24, 2009

To whom it may concern:

Enclosed please find the revision to our manuscript.

We sincerely appreciate the assistance of the editor as well as the helpful comments of the reviewers. In attempting to address all of the concerns of the reviewers, we have substantially revised our manuscript including re-focusing both the text (including the title) and analyses. We have removed almost 1,000 words from the original text, and have revised and shortened the tables to 3 (from 8).

We hope these changes have improved our manuscript. We look forward to your comments.

Sincerely,
Stephanie J. Frisbee.